Author's response to reviews

Title: Systemic autoimmune disease in asbestosis rapidly responding to anti-interleukin-1beta antibody canakinumab

Authors:

Laura Niccoli (lniccoli64@gmail.com)
Emanuele Cassarà (emanuele.cassar@gmail.com)
Olga Kaloudi (olgakaloudi@hotmail.com)
Carlotta Nannini (nannini.car@gmail.com)
Micaela Romagnoli (m.romagnoli@ausl.fo.it)
Fabrizio Cantini (fbrzcantini@gmail.com)

Version: 2
Date: 10 November 2014

Author's response to reviews:

Title page: the e-mail addresses of all authors were added.
Authors' details: Laura Niccoli MD, Consultant in Rheumatology, Rheumatology Division, Hospital of Prato, Italy. E-mail: lniccoli64@gmail.com; Emanuele Cassarà MD, Consultant in Rheumatology, Rheumatology Division, Hospital of Prato, Italy. E-mail: emanuele.cassar@gmail.com; Olga Kaloudi MD, Consultant in Rheumatology, Rheumatology Division, Hospital of Prato, Italy. E-mail: olgakaloudi@hotmail.com; Carlotta Nannini MD, Consultant in Rheumatology, Rheumatology Division, Hospital of Prato, Italy. E-mail: nannini.car@gmail.com; Micaela Romagnoli MD, Consultant in Pneumology, Pneumologie et Addictologie CHU Arnaud de Villeneuve 191, Av. du Doyen Gaston Giraud 34295 Montpellier, France. E-mail: m.romagnoli@ausl.fo.it; Fabrizio Cantini Professor. PhD, Consultant in Rheumatology, Director, Rheumatology Division, Hospital of Prato, Italy. E-mail: fbrzcantini@gmail.com

Submitting and corresponding Author: Prof. Fabrizio Cantini MD, Rheumatology Division, Hospital of Prato, Piazza Ospedale, 1, 59100 Prato, Italy. Tel: +39 0574 807578; e-mail: fbrzcantini@gmail.com

Abstract. Line 45. 4 instead of 3.

Page 2. Lines 23 to 36. The sections related to patient' consent, competing interests, authors' contributions, and acknowledgements were included.

Main text. Abbreviations were included and the list of abbreviations was indicated at lines 165 to 170.

Main text. Lines 114, 115. The following sentence was added: The therapy was well tolerated without adverse events or side effects.

Lines 163. The timeline was included in the manuscript.

The page and line numbering was added to the manuscript.

An additional file containing the Case report checklist was included in the
submission.